Author's response to reviews

Title: Selection of two emerging pathogens, Brevundimonas diminuta and Ochrobactrum anthropi, in a cystic fibrosis patient treated with colistin

Authors:

Magalie Menuet (magalie.menuet@wanadoo.fr)
Fadi Bittar (bittar_fadi@hotmail.com)
Nathalie Stremler (nathalie.stremler@ap-hm.fr)
Jean-Christophe Dubus (jean-christophe.dubus@ap-hm.fr)
Jacques Sarles (jacques.sarles@ap-hm.fr)
Didier Raoult (didier.raoult@medecine.univ-mrs.fr)
Jean-Marc Rolain (jm.rolain@medecine.univ-mrs.fr)

Version: 2 Date: 11 October 2007

Author's response to reviews:

Dear Editor,

Please find enclosed a manuscript entitled « Selection of two emerging pathogens, Brevundimonas diminuta and Ochrobactrum anthropi, in a cystic fibrosis patient treated with colistin ». This case report reemphasizes the emergence and selection of colistin-resistant bacteria in such patients. We believe this could be of interest for your readers. I hope it could be published as a case report in Journal of Medical Case Report.

All co-authors have seen and approved the manuscript and have complied with the stipulations in the Instructions to authors. No conflict of interest.

As potential unbiased reviewers we propose:

G. greub: gilbert.greub@chuv.ch
JL Mainardi: jlmainar@bhdc.jussieu.fr

Sincerely yours.

Pr ROLAIN Jean-Marc